Author’s response to reviews

Title: Gastroprotective effect of Desmosdumotin C isolated from Mitrella Kentii against ethanol-induced gastric mucosal hemorrhage in rats: possible involvement of glutathione, heat-shock protein-70, sulphydryl compounds, nitric oxide and anti-helicobacter pylori activity

Authors:

Heyam M A Sidahmed (diamondhm@hotmail.com)
Ainnul Azizan (ainnol_azizan@yahoo.com)
Syam Mohan (syammohan@yahoo.com)
Mahmood A Abdulla (mahmood955@yahoo.com)
Siddig I Abdelwahab (Siddigroa@yahoo.com)
Manal M E Taha (manalroa@yahoo.com)
A. Hamid A. Hadi (ahamid@um.edu.my.com)
Kamal A Ketuly (KKetuly@yahoo.com)
Najihah M Hashim (najihahmh@um.edu.my)
Mun F Loke (lokemunfai@um.edu.my)
Jamuna Vadivelu (jamuna@um.edu.my)

Version: 2 Date: 20 November 2012

Author’s response to reviews:

Not applicable since this is the initial submission